\newcommand{\noopsort}[1]{} \newcommand{\printfirst}[2]{#1}
  \newcommand{\singleletter}[1]{#1} \newcommand{\switchargs}[2]{#2#1}
\begin{thebibliography}{10}

\bibitem{attributor}
{Attributor - Subscription based web monitoring platform for content reuse
  detection}.
\newblock {http://www.attributor.com}.

\bibitem{berne}
{Berne Convention for the Protection of Literary and Artistic Works}.
\newblock {http://law-ref.org/BERN/index.html}.

\bibitem{n3logic}
Tim Berners-Lee, Dan Connolly, Lalana Kagal, Yosi Scharf, and Jim Hendler.
\newblock N3logic: A logical framework for the world wide web.
\newblock {\em Journal of Theory and Practice of Logic Programming}, 2007.

\bibitem{bliptv}
{blip.tv - Hosting, distribution and advertising platform for creators of web
  shows}.
\newblock {http://blip.tv/�}.

\bibitem{DBLP:conf/esws/BonattiDFNOPS06}
Piero~A. Bonatti, Claudiu Duma, Norbert~E. Fuchs, Wolfgang Nejdl, Daniel
  Olmedilla, Joachim Peer, and Nahid Shahmehri.
\newblock Semantic web policies - a discussion of requirements and research
  issues.
\newblock In {\em ESWC}, pages 712--724, 2006.

\bibitem{browser-market}
{Browser Market Share - 2009}.
\newblock {http://marketshare.hitslink.com/browser-market-share.aspx?qprid=0}.

\bibitem{by-legal-code}
{Creative Commons BY 3.0 Unported Legal Code}.
\newblock {http://creativecommons.org/licenses/by/3.0/legalcode}.

\bibitem{google_cc_search}
{Creative Commons Customized Search in Google}.
\newblock {Google Advanced Search Enables CC-Customized Searching,
  http://creativecommons.org/press-releases/entry/5692}.

\bibitem{liblicense}
{Creative Commons LibLicense library which integrates license metadata with
  content}.
\newblock {http://wiki.creativecommons.org/Liblicense}.

\bibitem{license-properites}
{Creative Commons License Properties}.
\newblock {License Properties,
  http://wiki.creativecommons.org/License-Properties}.

\bibitem{nc_user_study}
{Creative Commons Noncommercial study interim report,
  http://mirrors.creativecommons.org/nc-study/NC-Use-Study-Interim-Report-2009%
0501.pdf}.
\newblock {some title}.

\bibitem{copyright_debate}
{Debate on copyright and wrongs: "Existing copyright laws do more harm than
  good"}.
\newblock {http://www.economist.com/debate/days/view/310}.

\bibitem{license_babel}
{Different Open source Licenses}.
\newblock {http://www.opensource.org/licenses/alphabetical}.

\bibitem{exif}
{Exchangeable Image File Format }.
\newblock {http://www.exif.org/specifications.html}.

\bibitem{DBLP:conf/ccs/FeigenbaumFSS01}
Joan Feigenbaum, Michael~J. Freedman, Tomas Sander, and Adam Shostack.
\newblock Privacy engineering for digital rights management systems.
\newblock In Tomas Sander, editor, {\em Digital Rights Management Workshop},
  volume 2320 of {\em Lecture Notes in Computer Science}, pages 76--105.
  Springer, 2001.

\bibitem{xul}
{Firefox XUL Documentation}.
\newblock {https://developer.mozilla.org/En/XUL}.

\bibitem{flickrapi}
{Flickr API}.
\newblock {http://www.flickr.com/services/api}.

\bibitem{flickr_url}
{Flickr Farm URL}.
\newblock {http://www.flickr.com/services/api/misc.urls.html}.

\bibitem{flickroo}
{Flickr image reuse for openoffice.org}.
\newblock
  {http://wiki.creativecommons.org/Flickr-Image-Re-Use-for-OpenOffice.org}.

\bibitem{flickrlib}
{FlickrLib - Python wrapper for the Flickr API}.
\newblock {http://monotonous.org/2005/11/26/flickrlib-05/}.

\bibitem{hal08cc}
{Hal Abelson, Ben Adida, Mike Linksvayer, Nathan Yergler}.
\newblock {ccREL: The Creative Commons Rights Expression Language}.
\newblock {\em Creative Commons Wiki}, 2008.

\bibitem{DBLP:conf/policy/HansonBKSW07}
Chris Hanson, Tim Berners-Lee, Lalana Kagal, Gerald~J. Sussman, and Daniel~J.
  Weitzner.
\newblock Data-purpose algebra: Modeling data usage policies.
\newblock In {\em POLICY}, pages 173--177. IEEE Computer Society, 2007.

\bibitem{credit-on-flickr}
{How to attribute Flickr images}.
\newblock {http://www.squidoo.com/cc-flickr\#module12311035}.

\bibitem{ccval}
{Hugo Dworak, Creative Commons License Validation Service}.
\newblock {http://validator.creativecommons.org/}.

\bibitem{iptc}
{International Press Telecommunications Council Photo Metadata Format}.
\newblock {http://www.iptc.org/IPTC4XMP/}.

\bibitem{dist}
{Ivan Herman}.
\newblock {RDFa Distiller}, 2008.

\bibitem{harvey-thesis}
Harvey~C Jones.
\newblock Xhtml documents with inline, policy-aware provenance.
\newblock Master's thesis, Massachusetts Institute of Technology, May 2007.

\bibitem{jquery}
{JQuery Javascript Library by John Resig}.
\newblock {http://jquery.com}.

\bibitem{json}
{JSON - JavaScript Object Notation}.
\newblock {http://www.json.org/}.

\bibitem{air}
Lalana Kagal, Chris Hanson, and Daniel Weitzner.
\newblock Using dependency tracking to provide explanations for policy
  management.
\newblock {\em IEEE Policy 2008}, 2008.

\bibitem{attributor-prob}
{Ken Doctor, Blog Entry on "Attributor Fair Syndication Consortium Completes
  Newspaper Trifecta" }.
\newblock
  {http://www.contentbridges.com/2009/04/attributor-ad-push-on-piracy-complete%
s-newspaper-trifecta.html}.

\bibitem{qsign}
Jong~Wook Kim, K.~Sel�uk Candan, and Junichi Tatemura.
\newblock Efficient overlap and content reuse detection in blogs and online
  news articles.
\newblock In {\em 18th International World Wide Web Conference (WWW2009)},
  April 2009.

\bibitem{licensetagger}
{License Tagger}.
\newblock {http://wiki.creativecommons.org/Licensetagger}.

\bibitem{mozcc}
{MozCC - Firefox extension to discover Creative Commons licenses}.
\newblock {http://wiki.creativecommons.org/MozCC}.

\bibitem{newscredit}
{News Credit - WordPress plugin}.
\newblock {http://newscredit.org}.

\bibitem{operator}
{Operator Firefox Addon which discovers Microformats and RDFa in web pages}.
\newblock {https://addons.mozilla.org/en-US/firefox/addon/4106}.

\bibitem{owlmusicsearch}
{OWL Music Search}.
\newblock {http://www.owlmusicsearch.com/��}.

\bibitem{picscout}
{picScout - Image tracker for stock photography agencies and professional
  photographers}.
\newblock {http://www.picscout.com}.

\bibitem{powder}
{Protocol for Web Description Resources (POWDER)}.
\newblock {http://www.w3.org/2007/powder/}.

\bibitem{qdos}
{QDOS, Product developed by Garlik to measure the online status of an
  individual}.
\newblock {http://qdos.com/}.

\bibitem{rdf}
{RDF - Resource Description Framework}.
\newblock {http://www.w3.org/TR/rdf-syntax-grammar/}.

\bibitem{rdfa}
{RDFa - Resource Description Framework in Attributes}.
\newblock {http://www.w3.org/2006/07/SWD/RDFa/syntax/}.

\bibitem{scam}
N.~Shivakumar and H.~Garcia-Molina.
\newblock Scam: A copy detection mechanism for digital documents.
\newblock In {\em Second Annual Conference on the Theory and Practice of
  Digital Libraries}, 1995.

\bibitem{spinxpress}
{SpinXpress - Collaborative media production platform}.
\newblock {http://spinxpress.com}.

\bibitem{technorati}
{Technorati API}.
\newblock {http://technorati.com/developers/api/}.

\bibitem{thinkfree}
{Think Free - Java based web office suite}.
\newblock {http://www.thinkfree.com/}.

\bibitem{tab}
{Tim Berners-Lee}.
\newblock {Tabulator Redux: Browing and Writing Linked Data }.
\newblock In {\em {Linked Data on the Web Workshop at WWW08}}, 2008.

\bibitem{mozclip}
{Using the Mozilla Clipboard, Mozilla Developer Center}.
\newblock {https://developer.mozilla.org/en/Using-the-Clipboard}.

\bibitem{info-account}
Daniel~J. Weitzner, Harold Abelson, Tim Berners-Lee, Joan Feigenbaum, James
  Hendler, and Gerald~Jay Sussman.
\newblock Information accountability.
\newblock {\em Communications of the ACM}, June 2008.

\bibitem{danny_info_account}
Daniel~J. Weitzner, Harold Abelson, Tim Berners-Lee, Chris Hanson, James
  Hendler, Lalana Kagal, Deborah~L. McGuinness, Gerald~Jay Sussman, and
  K.~Krasnow Waterman.
\newblock Transparent accountable data mining: New strategies for privacy
  protection.
\newblock Technical Report MIT-CSAIL-TR-2006-007, Massachusetts Institute of
  Technology Computer Science and Artificial Intelligence Laboratory, January
  2006.

\bibitem{picscout-prob}
{William Faulkner, Tales From The IT Side: "PicScout, Getty Images and Goodbye
  iStockPhoto..!"}.
\newblock
  {http://williamfaulkner.co.uk/wordpress/2007/09/picscout-getty-images-and-go%
odbye-istockphoto/}.

\bibitem{xmp}
{XMP - Extensible Metadata Platform}.
\newblock {http://www.adobe.com/products/xmp/index.html}.

\bibitem{yahoo_cc_search}
{Yahoo Creative Commons Search}.
\newblock {http://search.yahoo.com/cc}.

\end{thebibliography}

\chapter{Summary}
\label{summary}

\section{Contributions}

We have provided an assessment on the number of license violations on the Web, and thus highlighted the need to make Web users more aware of the licensing options available to them. The tools we have implemented provide policy awareness using existing Semantic Web technologies. These tools also provides a platform to use the data exposed on the Semantic Web. 

The tools are:

\begin{enumerate}

\item \emph{Creative Commons License Violations Validator for Flickr Images}: 

This validator can be used to check if a certain Web site has used content inappropriately and thus violated anybody else's license terms when reusing Flickr images on the Web. If there are any violations, the validator provides the necessary information to include to make it license-compliant.

\item \emph{Semantic Clipboard}:

This detects reusable images while a user is browsing and enables them to seamlessly integrate those images along with their metadata. Such addition of license metadata enables the transparent transfer of content between applications, and makes people more aware of the policies associated with content reuse. 

\end{enumerate}

\section{Challenges}

There are several limitations in the software we have developed for this project. Some of them are due to the lack of clear-cut definitions of license terms that can be directly transcoded to programmable verification methods. Some are due to portability problems. Some are due to user practices, while some are due to wide adoption of certain technologies. These challenges, and the possible future directions to circumvent those are given in the following sections.

\subsection{Challenges for the Attributions License Violations Validator}

\subsubsection{Tracking Provenance}

The validator sufficiently addresses the problem of inappropriate content reuse, specifically Flickr image reuse, on the Web. However we are unable to guarantee that it will handle all the cases. A malicious user could  very easily change the image URI by uploading the same image on a different server. The license violations detection can only work if the the image URI is actually linked from the Flickr site. 

Another interesting scenario is that any uploader can assign CC licenses to images on Flickr regardless of that user having the actual rights to do so. In other words, if someone uploads a copyrighted photo from \emph{Getty Images} and assigns a CC license on Flickr, and an innocent user downloads and uses this photo, then that user will be violating copyright law without the user knowing it. 

Therefore, we need to have some capability to track provenance of image data, and be able to identify whether a particular image has been used elsewhere in a manner that violates the original license terms. There are two hurdles in achieving that. \emph{First,} image transformations are very easy to perform with image manipulation programs available today. Hence, even if there is a giant database of images (perhaps implemented by indexing the fingerprints of all the images on the Web), if one chooses to cheat, it can be easily done by changing a small unnoticeable pixel in the image and changing the fingerprint as it is non-invariant to image transformations. Therefore, we are assuming that this tool will be used in a  non-adversarial manner by honest people who wish to validate their documents against attributions license violations. \emph{Second}, provenance might mean different things to different people. For example, one might think of provenance preservation as linking to the original URI, without embedding it in some other location. However, one might also argue that provenance would be preserved if the content is embedded in a new location, but mentions the original URI from where it came from. In the validator we have implemented, we assume that provenance is preserved by linking to the original image URI from Flickr. 

\subsubsection{Subsequent Changes in the License}

There is no law preventing a user from changing the license given to an image at any point in time. This can even be from a CC license to copyright protected. This makes the reuser vulnerable to copyright infringement claims by the original owner, putting the burden of proof on the reuser's shoulders. Therefore, a mechanism to record temporal changes in the licensing information will be very useful.

\subsubsection{Scope of the Human Readable Attribution Notice}

One of the major assumptions we have made in developing this tool is that attribution to be specified within the parent node or the sibling nodes of the containing image element. Otherwise we classify it an instance of misattribution. This assumption works practically and seems to be the most logical thing to do. However, since there is no standard agreement as to what the correct scoping for attribution is, this assumption can give a wrong validation result. The solution to this problem can be in two folds. (1) CC should give a guideline as to what the correct scoping of attribution should be relative to the content that is attributed. (2) Flickr (any other such service) should expose the license metadata as RDF, instead of providing an API to query with. The latter method is preferred as it enables data interoperability and relieves the tool authors from having to write data wrappers  for each service.

\subsubsection{Limited Flickr Support for License Expression}

Flickr is still using the older CC 2.0 recommendation which specifies that the license metadata be included in HTML comments. This makes it impossible to process the metadata programmatically. This recommendation has been superseded by ccREL \cite{hal08cc} that was published in March 2009. But Flickr has not yet implemented this recommendation, and as a result, users are not able to express \emph{CC-Zero} (which means no rights are reserved with the image), or specify additional use restrictions via the \emph{cc:morePermissions}, or use many of the other numerous additions in the new recommendation that supports richer semantics including the \emph{cc:attributionName} and \emph{cc:attributionURL}. 

% The following paragraph is disputable - as I have seen evidence otherwise
%Also, Flickr does not allow enough granularity when specifying the licenses. In other words, users cannot control their rights on the individual photos or on the photo sets. The license they specify will be applied to their entire collection of photos. It would be useful if users can specify the licenses selectively. 

\subsection{Challenges for the Semantic Clipboard}

\subsubsection{License Granularity in the HTML DOM}

Content creators have the freedom to assign whatever the license they see fit to their works. However, there is no standard agreement as to the granularity to which the license is applied. For example the decision about whether to apply the license to the entire document where the content is found, or whether to apply it to the individual content item is not specifically stated in the ccREL. Semantic Clipboard assumes that a license is applied to each image, and extracts the RDF metadata for that image in order to ascertain what the license for that particular image is. This assumption holds if the original content creator uses a tool when publishing her work that will automatically embed the license metadata for each content item rather than giving an all-encompassing license at the page level. 

\subsubsection{Browser Dependence}

One of the major drawbacks of the Semantic Clipboard is that it is Firefox browser-dependent. Firefox has only 22.48\% of the browser market share according to \cite{browser-market}. Therefore, it seems that this application will not be widely adopted if it is not made available in all of the other major browsers. Developing an Opera Widget, a Chrome Extension, a Safari Plugin, an Internet Explorer Content Extension or completely making this tool browser independent seem to be a viable future direction of the project in terms of making the tool available to the masses.

\subsubsection{Usability vs. Operating System Independence}

It would be convenient to copy the attribution XHTML as \emph{rich text} into a text editor which accepts hyperlinked text and displays it without the source text. Including this feature would be useful from a usability perspective, but it would raise concerns on the portability of the Semantic Clipboard across different operating systems. The Semantic Clipboard needs to have knowledge of the allowed data formats in the target applications. But this is a very operating system dependent parameter. The \emph{data flavor} used when copying an image in the current implementation is `ASCII text'. Thus the application is portable to any Operating System. We have only tested the application on Mac OS X 10.5.6 and Ubuntu 9.04, but we assume that it should work in other platforms. 

\section{Future Work}

\subsection{Check for Other Types of License Violations}

Detecting whether an image has been used for any commercial use would be of much interest to content creators, especially if the second use of the image decreases the monetary value of the original image. But as was discussed in Chapter \ref{background}, from the point of view of the content consumers, it is hard to give a precise definition as to what constitutes a \emph{Commercial Use}. For example, if the images are used in a Web site that has subscribed to a dynamic advertisement generation service, it can be easily argued that the advertisements that appear on the page and the image content that is embedded in the page have no direct correlation. However, as discovered by the CC survey \cite{nc_user_study}, large number of content creators actually generate revenue indirectly through advertisements. Therefore, if the definition of \emph{non commercial use} becomes clearer and much more objective, the validator can be used to check for such violations as well.
%Since certain Web sites can be white-listed as non-profit and non-commercial, perhaps based on the specific top level domain name such as `.org' associated with the Web site, the tool can make the inference that embedding an image in such a Web site would not lead to a non-commercial license violation. 

It would also be interesting to check for share-alike license violations. These violations happen when a conflicting license is given when the content is reused. The solution, therefore, is to check the RDFa in both the original page and in the page where the image was embedded to see if the latter is the same as the original CC license. 

\subsection{Give Credit to the Original Content Creator as Requested}

%In the world of digital data, licenses are utilized to grant certain rights to the user by actually reducing the rights held by the user. For example, data created by user X are protected for X's use completely and fully by the copyright law. However, X can grant a license to use the data for Y's work provided that Y use it for non-commercial purposes. If that person start using X's data for commercial purpose, X can revoke the license by asking Y to stop using the data. X can further obligate the Y, to do something when using X's data Ñ for example, X can ask Y to pay X some money, or give X the credit the way X wants it. 

The requirement for attribution is a form of social compensation for reusing one's work. While the mentioning of one's name or the URI when attributing trades attention to an individual, other forms of \emph{attention mechanisms} can also be implemented. For example, a content creator can obligate the users of her works to give monetary compensation or require that they include certain ad links in the attribution XHTML or give attribution in an entirely arbitrary manner. These extra license conditions can be specified using the \emph{cc:morePermissions}. Tools could be built to interpret these conditions and give credit to the original creator as requested.

\subsection{Extend to Other Media Types}

We have only explored one domain of content, specifically image reuse on the Web. However, there are billions of videos uploaded on YouTube, and potentially countless number of documents on the Web, which have various types of licenses applied. While organizations such as Mobile Picture Association of America (MPAA), Recording Industry Association of America (RIAA) and other such big organizations are working towards preserving the rights of the works of their artists on YouTube, other video and audio sharing sites and peer-to-peer file sharing networks, there are no viable alternatives for ordinary users who intend to protect their rights using CC. Thus a solution of this nature which detects CC license violations based on the metadata of free-floating content will be very useful. 

\subsection{License Granularity}

The tool we have developed only works if every image found on the page has it's own license. Possible extensions of this tool would be to determine the license of an image when it does not have a license of it's own, but is contained within a page that has a license or is a member of a set of images (e.g. a photo album) that has a license. Protocol for Web Description Resources (POWDER) \cite{powder}, a mechanism that allows the provision of descriptions for groups of online resources, seems like a viable method to making the license descriptions about the resources explicit. Tool builders can then rely on the POWDER descriptions to help users to make appropriate content reuse decisions. 

 %From a practical standpoint, it can be assumed that only one license is applied to the entire page or a set of images. 

\subsection{Persistent Data Storage}

Currently, images that are copied with their metadata to the Semantic Clipboard are overwritten when new content is copied to the clipboard. In other words, the tool only supports copying of one image at a time. It would be useful to have a persistent data storage to register content along with their license metadata, index them, make them persistent across browser sessions and use it whenever the user needs it.

\subsection{User Study}

It would be interesting to measure how user behavior changes with the introduction of tools such as the License Violations Validator and the Semantic Clipboard. A measurement of the increased (or decreased or unchanging) level of license awareness would be an important metric in determining the success of these tools. Therefore, we would like to perform a controlled user study to determine how many violations a set of users make before and after the introduction of these tools during a limited period of time.

\subsection{Widening Social Criteria}

The central idea presented in this thesis is not limited to Creative Commons licenses. One could also combine it with a reasoning engine that reasons over security clearances to produce policy-aware tools that can calculate the appropriate level of data classification. Similarly, one could combine it with a tool that is aware of assertions regarding data reuse and create a tool that checks for violations of citizens' privacy \cite{danny_info_account}. Therefore, we hope the work on this thesis will spur research on how best to preserve data provenance in other domains.


\section{Conclusion}

As the license violations experiment indicated, there is a strong lack of awareness of licensing terms among content reusers. This raises the question as to whether the machine readable licenses are actually working. Perhaps more effort is needed to bring these technologies to the masses, and more tools are needed to bridge the gap between the license-aware and the license-unaware. An important research question that stems from this work is the method of  provenance preservation of content on the Web. We have trivially assumed URIs as the provenance preservation mechanism when developing the tools described in this thesis. However, it would be an interesting challenge to track provenance based on the content itself, without having to rely on a unique identifier. This would enable us to find out license violators easily, in addition to validating one's own work for any violations. Also, programmatically determining whether a particular reuse of material is allowable or not is subjective, especially since some of the laws and standards have been quite ambiguous in defining these terms. 

%Another question which is at a much more philosophical level is whether every resource on the web be licensed and that fact be made aware of? We have demonstrated that it is possible to do so with tangible media such as images and text. We can easily extend to other types such as video and audio as well. However what about intangible intellectual property such as ideas expressed on the Web? 

In general, social constraints are functions of any part of the blossoming Social Web we are experiencing today. As we are living in an era of increasing user generated content, these constraints can be used to communicate the acceptable uses of such content. We need tools, techniques and standards that strike an appropriate balance between the rights of the originator and the power of reuse. The rights of the originator can be preserved by expressing what constitutes appropriate uses and restrictions using a rights expression language. These rights will be both machine and human readable. Reuse can be simplified by providing the necessary tools by leveraging these machine readable rights to make the users more aware of the license options available and ensure that the user be license or policy compliant. Such techniques can be incorporated in existing content publishing platforms or validators  or even Web servers to make the process seamless. This thesis has demonstrated several tools that lay the foundation for such policy aware systems. We hope these will stimulate research in this area in the future.





